Reviewer
The study is interesting, in special group of patients with mortality but it lacks the following items a- The effect of adding immune therapy of HCC, adding immune therapy for this group of patients significantly improved survival than other line of treatment and arm of group of patients who received immune therapy will improve the value of this study b- The cost analysis of the conservative therapy versus stenting of P c- The complication of invasive procedures.

Dear Reviewer

A. Among the included studies, None of them showed immune therapy of HCC.
B. Among the included studies, None of them showed the cost analysis of the conservative therapy versus stenting of P.
C. Among the included studies, all patients have no surgery-related complications such as upper gastrointestinal bleeding or liver failure.